Author's response to reviews

Title: The impact of the Cyprus comprehensive smoking ban on air quality and economic business of hospitality venues

Authors:

Costas A Christophi (costas.christophi@cut.ac.cy)
Martha Paisi (martha.paisi@cut.ac.cy)
Despina Pampaka (despina.pampaka@cut.ac.cy)
Martha Kehagias (kehagias@yahoo.com)
Constantine Vardavas (vardavas@hsph.harvard.edu)
Gregory N Connolly (gconnoll@hsph.harvard.edu)

Version: 2 Date: 1 August 2012

Author's response to reviews:

In response to the Journal Editorial Office's comment, regarding ethical approval, we added the sentence "The study was exempt from any ethical review process as there were no human subjects recruited and no personal data collected." in the Methods section.